Author's response to reviews

Title: Prognostic value of alveolar volume in systolic heart failure: a prospective observational study

Authors:

- Massimo Miniati (massimo.miniati@unifi.it)
- Simonetta Monti (monti@ifc.cnr.it)
- Matteo Bottai (matteo.bottai@ki.se)
- Ivana Pavlickova (iva.pav@tiscali.cz)
- Claudio Passino (passino@ftgm.it)
- Michele Emdin (emdin@ftgm.it)
- Roberta Poletti (poletti@ftgm.it)

Version: 2 Date: 30 June 2013

Author's response to reviews:

To the Editors,

BMC Pulmonary Medicine

Dear Sirs,

please find attached the revised version of the manuscript entitled “Prognostic value of alveolar volume in systolic heart failure: a prospective observational study”.

A point-by-point response to the reviewers’ comments is also attached.

The abstract has been slightly modified according to reviewers’ suggestions.

Similarly, figure 1 has been modified based on reviewers’ suggestions.

The Introduction, Results, and Discussion contain new paragraphs that are marked in red in the revised version of the manuscript.

Two references were added that are marked in red.

Very obliged,

Massimo Miniati, MD, PhD
Associate Professor of Internal Medicine,
University of Florence, Italy.